Author's response to reviews

Title: SOCS3 inhibiting migration of A549 cells correlates with PYK2 signaling in vitro.

Authors:

Siyang Zhang (zhangsiyang_cmu11032@126.com)
Dawei Guo (gdw80@163.com)
Lili Jiang (neko200321063@126.com)
Qingfu Zhang (qingfu1981@126.com)
Xueshan Qiu (qiuxueshan_pathology@126.com)
Enhua Wang (wangelo@hotmail.com)

Version: 8 Date: 22 May 2008

Author's response to reviews: see over
Dear Editors,

Thank you for your letter. We are very pleased to learn that our manuscript has been accepted for publication in BMC Cancer with some more formatting revisions. We have completed the revised manuscript according to the editors’ recommendations, and checked the manuscript carefully. We greatly appreciate your help concerning improvement to this manuscript. Moreover, we’ll keep the manuscript being revised until you’re satisfied with it. Thank you very much! If there are any more problems, please contact me without hesitation.

Sincerely yours,

Xueshan Qiu

The point-by-point response to the editors is shown as follows:

Figure order- Figures have been cited in ascending numerical order in the revised text.

The original figure 4 is changed to figure 1 in the text, and the original figure 1,2,3 are changed to 2,3,4, respectively.

Figure 7 is currently mentioned after figure 5 and 6 in the revised text as follows:

“As shown in Fig.7, exogenous SOCS3-SH2, -KIR and -box mutants were detected in A549 cells, as indicated by the myc tag”, which was inserted in the last paragraph of the Results section.